Author’s response to reviews

Title: Increased risk of severe congenital heart defects in offspring exposed to selective serotonin-reuptake inhibitors in early pregnancy - an epidemiological study using validated EUROCAT data

Authors:

Tanja M Knudsen (tan.knudsen@hotmail.com)
Anne V Hansen (avih@sund.ku.dk)
Ester Garne (egarne@dadlnet.dk)
Anne-Marie N Andersen (amny@sund.ku.dk)

Version: 5 Date: 26 August 2014

Author’s response to reviews:

Dear editor

Thank you for your comment and the interest you have shown our research article. We have addressed the comment in a revised manuscript. Below the changes we have made are described giving a point-by-point response to the concern.

Editorial Comments

1. STROBE

Please include a statement in the methods section of your manuscript confirming that your research has adhered to the STROBE guidelines for observational studies as outlined here: http://www.strobe-statement.org.

Our response:

Thank you for the feedback on the method. We have included a statement in the method section, which confirms that our research adheres to the STROBE guidelines for observational studies.

In order to meet the STROBE guidelines we have added a phrase about external validity in the discussion section.

We look forward to hearing from you again.

Best regards
Tanja Majbrit Knudsen